Author’s response to reviews

Title: The prevalence of heterozygous F12 mutations in Chinese population and its relevance to incidents of thrombosis

Authors:

Xi Wu (wuxi_0723@163.com)
Qiulan Ding (qiulan_ding@shsmu.edu.cn)
Xuefeng Wang (wxf63@shsmu.edu.cn)
Jing Dai (wzdjsx7675@hotmail.com)
WENMAN WU (wenmanwu@shsmu.edu.cn)

Version: 3 Date: 26 Feb 2018

Author’s response to reviews:

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication (e.g. "Backgournd").

The manuscript was edited by a native English speaker and we hope the revision will meet the standard for publication.

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

The abbreviations have been listed as required.

3. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

The statement has been inserted
4. Please change the ages to age ranges in Table 1, in order to protect the anonymity of the patients (e.g. 20-29, 30-39, etc.)

The table has been changed as suggested.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.